Author's response to reviews

Title: Arm rotated medially with supination - the ARMS deformity: description of its surgical correction

Authors:

Rahul K Nath (drnath@drnathmedical.com)
Chandra Somasundaram (chandra@drnathmedical.com)
Sonya E Melcher (sonya@drnathmedical.com)
Meera Bala (meera@drnathmedical.com)
Melissa J Wentz (melissa@drnathmedical.com)

Version: 7 Date: 11 March 2009

Author's response to reviews:

Formatting changes have been made as requested. Figure 8 will be removed. It was removed from the text during the revision process, but we neglected to remove the image from the system.